Author's response to reviews

**Title:** In vitro tooth whitening effect of two medicated chewing gums compared to a whitening gum and saliva

**Authors:**

Michael Moore (moorem@ipfw.edu)  
Nathalie Hasler-Nguyen (nathalie.hasler-nguyen@novartis.com)  
Geoffrey Saroea (geoffrey.saroea@novartis.com)

**Version:** 11  
**Date:** 8 August 2008

**Author's response to reviews:**

In table 1 and 2 numbers were placed under the title or column